Author's response to reviews

Title: Serum Immunoglobulin G, M and A Response to Cryptosporidium parvum in Cryptosporidium-HIV Co-infected Patients

Authors:

Kirti Kaushik (kaushikkirti@gmail.com)
Sumeeta Khurana (sumeetakhurana@hotmail.com)
Ajay Wanchu (awanchu@yahoo.com)
Nancy Malla (drmallanancy@gmail.com)

Version: 3 Date: 9 July 2009

Author's response to reviews: see over
To,
The Editor
BMC Infectious Diseases

Subject: Revised manuscript for publication in BMC infectious Diseases

Dear Sir,

Please find the revised manuscript entitled ‘Serum Immunoglobulin G, M and A Response to Cryptosporidium parvum in Cryptosporidium-HIV Co-infected Patients’ and pointwise reply to the reviewer’s comments.

Reply to Reviewer’s comments

We are thankful to the reviewers for their critical suggestions. We have modified the manuscript as per the suggestion of the reviewers.

Reviewer 1.

Points 2, 3 and 4: As per the suggestion of the reviewer a two-panel scattered plot of IgG (left panel) and IgA (right panel) OD values is included in the manuscript now where the X-axis contains the study groups (Group I, II, III, IV and V) and the Y-axis shows OD. The plot includes OD for test samples, positive and negative controls, cut-off for positivity and mean values for OD for each group (Fig 1). This is now included in lines 17-19 on page 10.

Thanks and regards,

Kirti Kaushik